Author's response to reviews

Title: Paired design study by real-time PCR: miR-378* and miR-145 are potent early diagnostic biomarkers of human colorectal cancer

Authors:

peng juan (dielianhuayaya@163.com)
xie zhengyong (pwkxiezy@163.com)
cheng liyang (chliyang2008@sina.com)
zhang yuxin (zhangyuxin-53825@163.com)
yu hongping (yhp268@163.com)
chen juyong (294821766@qq.com)
kang huixin (1367706921@qq.com)

Version: 9 Date: 8 February 2015

Author's response to reviews: see over
Cover letter

1. We have improved the style of written English
2. The figures have been removed in the paper
3. We edit the paper according to the style of journal